Author's response to reviews

Title: Can an Online Clinical Data Management service help in improving data collection and data quality in a developing country setting?

Authors:

Maarten A Wildeman (m.wildeman@nki.nl)
Jeroen Zandbergen (j.zandbergen@nki.nl)
Andrew Vincent (a.vincent@nki.nl)
Camelia Herdini (camelia_herdini@yahoo.com)
Jaap M Middeldorp (j.middeldorp@vumc.nl)
Renske Fles (r.fles@nki.nl)
Otilia Dalesio (o.dalesio@nki.nl)
Emile Donk (e.vd.donk@nki.nl)
I. B. Tan (i.tan@nki.nl)

Version: 4 Date: 26 July 2011

Author's response to reviews:

Dear Editor,

Thank you for re–reviewing article and allowing us the opportunity to resubmit our manuscript in line with the Trials Journal formatting guidelines.

Our modified manuscript now contains a Figures section, which we have included after the Reference list, listing titles and legends of the figures included in our manuscript.

After the Figures section we have added a Tables section, including all Tables referenced in our manuscript, each with their title and legend.

We hope our revisions answer to the Editors requests, thereby making our manuscript eligible for publication in your Journal.

Yours sincerely,

I. Bing Tan